BANNING US FOREIGN BRIBERY: DO US FIRMS WIN?

Roberto Ramos

CEMFI Working Paper No. 1309

September 2013

CEMFI Casado del Alisal 5; 28014 Madrid Tel. (34) 914 290 551 Fax (34) 914 291 056 Internet: www.cemfi.es

I thank Claudio Michelacci, Guillermo Caruana, David Dorn, Manuel Arellano, Samuel Bentolila, Stéphane Bonhomme, Eric Chaney, Julia Cagé, James Feigenbaum, Manuel Garcia-Santana, Saul Lach, Horacio Larreguy, Joan Llull, Michael Manove, Nathan Nunn, and seminar participants at CEMFI, SAAe Málaga 2011 and Harvard University Economic History Tea for very useful comments. I am also indebted to Jesús Saurina and Javier Delgado. Financial support of the Spanish Ministry of Education under research grant BES-2009-026803 is gratefully acknowledged. CEMFI Working Paper 1309 September 2013

BANNING US FOREIGN BRIBERY: DO US FIRMS WIN?

Abstract

I document a striking stock market underperformance of US firms in industries engaged in international trade during the period March 1976 to December 1977. I argue that a suitable candidate for explaining it is the US unilateral banning of foreign bribery that takes place in this period of time. I conduct a long-run event-study to analyse the stock market performance of firms differing in exposure to this policy. The results show that the cumulative abnormal returns of companies in industries most opened to international trade start to fall precisely the day banning foreign bribery is introduced in the political debate, and keep falling until it becomes law. I also show that the patterns observed are also evident when considering simply raw returns. Results are robust to different specifications and alternative explanations, such as oil prices or the exchange rate. I also disregard the influence of global shocks by performing the analysis for a country not subject to the policy. I conclude that the evidence points in the direction that the US unilateral banning of foreign bribery had a significant negative effect on the market value of firms most exposed to this policy.

Roberto Ramos Banco de España robertoramos8@gmail.com

1 Introduction

When Multinational Corporations do businesses abroad, they are often required to make questionable payments to foreign officials. Since the 1970s, a large stream of foreign corruption scandals has involved major companies in the world, showing that foreign bribery is prevalent and widespread when internationalized firms conduct businesses overseas.¹ Interestingly, these foreign payoffs were not illegal in any country in the world until the United States (US) prohibited them unilaterally in 1977, by means of the Foreign Corrupt Practices Act (FCPA). For the rest of the developed world, they remained within law. In 1997, the Organization of Economic Cooperation and Development (OECD) approved the Convention on Combating Bribery of Foreign Public Officials in International Business Transactions. Under this accord, signatory countries -representing three quarters of global exports- agreed to pass national legislation criminalizing the act of bribing foreign officials.

The US unilateral banning of foreign bribery offers an appealing experiment to measure how much value these payoffs had and how firms were affected by the policy banning them. In this paper, I provide an assessment of the consequences of the FCPA on US firms. I conduct an eventstudy analysis to gauge how the market value of companies responded to the FCPA. Under the assumption of efficiency in the financial markets, the stock market performance of firms around the time of the FCPA provides a measure of the value that the market gave to overseas payments. In principle, as foreign competition was not subject to the law, the one-sided prohibition put American Multinationals in disadvantage with respect to the rest of the world. This should translate in a decrease in the market value of firms affected by the policy.

The event-study I conduct has three major challenges. First, it is a long-run event study. The proposal, debate and approval of the FCPA took 21 months. As a consequence, as it is usual in the long-run event-study literature, I control for long-run common risk factors that explain realized returns. In particular, I estimate the Fama-French Three-Factor Model and compute abnormal returns as the part of the return not explained by the common factors. I follow closely the work by Abadie and Gardeazabal (2003).

Second, identifying the firms exposed to the legislation is not straightforward. In principle, companies that made use of foreign bribery or potentially could make use should be the firms most affected by the policy. However, by their own nature, bribes are unobserved. Therefore, I resort to indirect evidence. I choose exports as a proxy for exposure. Firms more engaged in international trade are in principle more likely to be affected. I use exports at the industry level due to data limitations and convenience for the analysis. I also explore whether results are robust to data on exports at the firm level.

And third, the 1970s is a turbulent period in the global economy, specially due to the two oil shocks of 1974 and 1979. For this reason, I make an effort to rigorously control for industry and global shocks that could provide alternative explanations to the observed returns. Aside from

¹See, for instance, the notorious cases of the US aerospace company Lockheed (1975), the German conglomerate Siemens (2006) and the UK defense contractor BAE Systems (2007).

excluding the oil firms from the analysis, I use industry dummies, expenditures in energy and the returns of firms in a country not affected by the FCPA -United Kingdom- to control for them.

I find a significant negative stock-market performance of firms in sectors engaged in international trade around the time of the proposal and approval of the FCPA. I interpret it -at least partly- as capturing the market effect of banning foreign bribery. For the firms in sectors most opened to international trade, these are of around 25%. I also use the Jensen's alpha approach, common in the long-run finance event-study literature, to provide further evidence. With this approach, the negative returns are of around 15%.

I use the event-study as motivating evidence, in order to study the data in a more systematic way. I run regressions of the firms' cumulative abnormal returns during the period of interest against the pre-policy exposure, controlling for equity characteristics that could help explain abnormal returns. I provide some alternative specifications on the benchmark portfolios and robustness checks. Overall, the qualitative results prevail.

Some papers have dealt previously with the economic consequences of the US banning of foreign bribery. Beck et al. (1991) shows that the US import share decreased in bribe-prone countries outside Latin America in the years after the FCPA was passed. This paper reversed the results of Graham (1984), who had found no difference on the evolution of the market share of the US industry between countries of high and low corruption. Hines (1995) examines the performance of the US businesses in the years after 1977. He finds negative effects on foreign direct investment (FDI), capital labour ratios, joint venture activity and aircraft exports. In 1981, in a survey of 250 large firms by the General Accounting Office, 32 per cent of companies reported a decrease in overseas business due to the FCPA, whereas 62 per cent thought that American companies could not compete against other abroad without the use of questionable payments.

In a closer study to this paper, Smith et al. (1984) show that firms that voluntarily reported foreign payoffs experienced negative abnormal returns at the date of the announcement. I also analalyse the abnormal returns of firms, but my focus is on the effect of the legislation that prohibited overseas payments. Therefore, I consider all firms listed in the New York Stock Exchange (NYSE), not only companies that self-reported foreign bribes. Also, my window of analysis is much wider, as I consider the whole period surrounding the anti-bribery legislation. This is important, as the process of enacting laws extends for several months and, therefore, the information content of the law -and hence the potential impact- is revealed slowly. By considering the complete timeline that led to the FCPA, we can better understand the *total* effect of the policy on the market value of firms.

In recent years, the research work on corruption has become much larger.² In particular, several papers have studied the rationale behind illegal payments and bribery. Svensson (2003) explains the variation in bribes across graft-reporting firms in Uganda, using variations in regulations across industries. Olken and Barron (2009) applies standard industrial organization theory to explain the corrupt behaviour of officials in Indonesia. Regarding corruption and international trade flows,

²For a recent survey, see Banerjee et al. (2009)

D'Souza (2009) examines how the OECD Anti-Bribery Convention shifted the trade flows towards low corrupt countries and Lambsdorff (1998) shows that the degree of corruption of importing countries has significant explanatory power for the trade structure of some exporting countries.

This paper analyses whether financial markets penalized firms engaged in international trade at the time of the approval of the anti-bribery legislation in the US. In this regard, as it tries to uncover the market value effects of questionable foreign payoffs, it adds to the literature of forensic economics, who explores the rationale behind illicit transactions. Examples include the value of political connections (Fisman (2001), Faccio (2006) and Fisman et al. (2006)) and the effect of conflict and wars on the market performance of firms (DellaVigna and La Ferrara (2009), Guidolin and La Ferrara (2007) and Dube et al. (2008)).

Finally, the event-study of this paper embeds in a large literature on the long-run event-study methodology. The papers of Campbell et al. (1997), Lyon et al. (1999) and Khotari and Warner (2007) are useful references. The event-study I conduct is closest to the spirit of Abadie and Gardeazabal (2003). They account for an overperformance of stocks of firms with significant activity in the Basque Country during a truce announced by terrorist group ETA, which has its origin in this Spanish region. Also related is the already cited work of Guidolin and La Ferrara (2007), who observe a significant drop on the price of stocks of firms with businesses in Angola, when an unexpected event drove the civil war affecting this country to a quick end.

The rest of the paper is organized as follows. Section 2 explains the event-study analysis. I provide a foundation for the exercise, detail the choice of the event-window, explain the empirical approach and present some graphical results. Section 3 provides a more systematic view of the results. I run regressions of the firms' cumulative abnormal returns against the exposure to the policy, controlling for industry shocks and other determinants of equity returns. I also provide evidence on the market value effect of the FCPA using a traditional method for calculating long-run abnormal returns, the so-called Jensen's alpha approach. Section 4 provides further evidence. I display the results of the same analysis on United Kingdom firms, which were not subject to the US policy. Also, I present a robustness check using firm-level export data. Section 5 concludes the paper.

2 Motivating Evidence: an Event-Study Analysis of the FCPA

In this section, I explain the the event-study analysis I conduct for the banning of foreign bribery in the US. I start by setting the rationale behind it, focusing on the specific characteristics that arise when trying to measure the effect of the FCPA. Then, I explain the choosing of the event-window, providing a description of the steps that lead the US to prohibit overseas payments. Finally, I present graphical results, who serve as motivating evidence for the next section.

2.1 The Event-Study Analysis

If financial markets are efficient, prices reflect all available information and therefore react only to changes in the business environment that affect expected profits. Consequently, if the FCPA put in disadvantage US firms, this should translate in a decrease in stock prices for companies exposed to the law. Note that this policy amounts to an increase in the cost of conducting businesses abroad. After implementation, companies must comply with formal bureaucracy dodged previously by means of illicit payoffs- or assume the risk of being fined if still resorting to questionable payments. Therefore, the stock market performance of firms during the period in which the FCPA was debated and approved provides a measure of the impact of prohibiting corporate foreign bribery. If banning foreign bribery is perceived as detrimental to firms engaged in international trade, companies exposed to it should underperform companies not exposed.

Contrary to more traditional event-studies, this one has three major challenges. First, it is a long-run event-study. By the own nature of the legislative process, subject to debate and voting in more than one House, the procedure of enacting laws spans several months in time. In particular, 21 months passed since the first bill regarding foreign bribery was proposed until the FCPA was approved. Therefore, the whole period of interest was comprised of small shocks affecting the bill and, as a result, the information content of the law was revealed slowly. As a consequence, it is necessary to look at how returns evolved during the whole period surrounding the FCPA in order to gauge properly the *total* effect of the law on the market value of firms.

The fact that the event-window is so wide makes necessary to control for long run risk factors in stock returns. I estimate the so-called Fama-French Three-Factor Model to account for this. In this regard, I follow closely Fama and French (1993) and Abadie and Gardeazabal (2003). I compute the abnormal returns of portfolios differing in the exposure to the policy during the event-window, controlling for long-run risk factors. In the next section, I also perform the so-called Jensen's alpha approach -or the calendar-time portfolio approach- to estimate the risk-adjusted abnormal performance, which is tradition in the finance long-run event-study literature.

Second, the classification of firms exposed to the policy in opposition to firms not exposed is not straightforward. In principle, companies exposed should be the ones that resorted -or potentially could have resorted- to foreign bribery. However, by their own nature, bribes are unobserved. As a consequence, I use indirect evidence. I take exports as a proxy for foreign operations and, therefore, as a proxy for the likelihood of resorting to foreign bribery.

And third, the 1970s are a turbulent decade in the global economy, specially due to the two oil shocks that hit it in 1974 and 1979. I make an effort to isolate the abnormal returns from these events. Although the specific time period around the FCPA is out of these oil shocks, during the whole analysis I exclude the oil firms, which nevertheless could be affected by expectations of higher future oil prices. I also control in the regressions for the cost of electricity and fuels at the industry level, as well as industry dummies to account for sectoral shocks. Finally, I also look at the stock market returns of a country not subject to the FCPA, the United Kingdom. Overall, these tests allow me to disregard global shocks at the industry level that could explain the observed returns.

In the next subsection, I explain the steps that lead the US to prohibit corporate foreign bribery, in order to specify the window of analysis of the event-study. Next, I define the proxy for exposure to the policy and present some facts about the data. Next, I present graphical results of the event-study.

2.2 Event Window: The US Foreign Corrupt Practices Act

The anti-bribery legislation in the US has its origin in the Watergate scandal in the early 1970s. The investigations that followed showed that some of the most important corporations of the country had made contributions to the 1972 President Nixon's reelection campaign, which were forbidden by law.

The fact that these corporations failed to report such contributions made the Securities and Exchange Commission (SEC) to step in. In 1974 it started an inquiry which lead to the revelation that the secret funds were spent not only on domestic political contributions but also on bribing officials in foreign countries.

In March 1975, the SEC moved against two major oil companies, Phillips Petroleum and Gulf Oil. The SEC alleged that these companies kept slush funds for political contributions and other purposes, although it did not say who received the funds. More importantly, on April 9, 1975, the Commission filed a complaint against United Brands Company for failing to disclose \$2 million in payments to Honduran officials and an unnamed European government, in order to secure favorable treatment. This marked the first foreign bribery scandal that hit the public opinion.

Importantly, bribing foreign officials -contrary to bribing domestic officials- was not against the US law at the time. Therefore, the SEC litigation was based on full and adequate disclosure of all financial information, as well as proper accounting.

In the months that followed the United Brands case, several other major US multinationals were accused by the Commsion of being involved in foreign bribery³. President Ford initially played down the scandals. However, in February 1976, facing growing pressure from society and the US Congress, which was conducting an inquiry, he condemned the payment of bribes overseas and began a review of ways to punish them.

The first legislative proposal regarding foreign bribery was introduced on March 11, 1976 by Senator William Proxmire, chairman of the Senate Banking Committee, who was probing the foreign bribery scandals since August 1975. The measure would make it a crime for US firms to bribe foreign officials, candidates and political parties. In June 1976, once it became clear that Congress would take action, President Ford presented his legislative proposal. He called for

³In June 1975, they were more than a dozen. By January 1976, 45 companies were involved. In May of that same year the SEC submitted a report to Congress on *Questionable and Illegal Corporate Payments and Practices*. It disclosed that 95 of the largest US corporations had admitted making questionable payments abroad. In total, more than 500 companies acknowledged having resorted to dubious foreign payoffs, including more than 100 listed on the Fortune 500. At least \$300 million in foreign payoffs were reported.

disclosure of all 'questionable payments', but he was not supportive of making them unlawful.

In September 1976 a bill making it illegal to bribe or attempting to bribe foreign officials under penalties of fines and jail was passed unanimously by the Senate. However, the adjournment of the US Congress for the Presidential Elections of November 1976 prevented the bill from getting through the House of Representatives and obtaining White House approval.

Shortly after the Congress convened in January 1977, Senators Proxmire and Williams introduced Senate Bill 305, similar in content as the one approved the previous year. In March 1977 the Carter administration gave its support to legislation making foreign bribery a criminal offence. This marked a change in the official US policy regarding foreign bribery. As Democratic Nominee, President Carter opposed Ford's legislative proposal on the grounds of being too lax.

Finally, after some amendments, Senate Bill 305 was approved with no opposition by the Senate and the House of Representatives. On December 19, 1977, President Carter signed the Foreign Corrupt Practices Act (FCPA) into law.

The law was divided in two sections -the anti bribery provisions and the accounting and internal control provisions. The anti-bribery provisions made it illegal for US businesses and persons to give anything of value to foreign officials, political parties or political candidates in order to influence their decisions. The accounting provisions obliged companies to keep books, records and accounts that accurately reflect the corporate payments and transactions.

Civil and criminal penalties imposed by the law applied both to individual and corporate violators. Persons violating the FCPA could be fined up to \$10,000 and served a 5-year prison sentence. Companies engaged in foreign bribery could be fined up to \$1 million.⁴ Fines imposed on individuals could not be paid by the companies.

Enforcement of the anti-bribery provisions was the primary responsibility of the Department of Justice. The SEC was mainly in charge of the accounting and internal control provisions. Within months of the approval of the law, Katy Industries Inc. was charged with violating the FCPA and a fine was imposed. Between 1977 and 1995, near twenty companies were charged under the FCPA. Corporate fines ranged between \$ 10,000 and \$3.45 million.⁵ In January 1995, Lockheed agreed to pay a record criminal fine of \$21.8 million and one official served a prison sentence. Recent penalties involve multinationals such as Siemens (\$800 million in 2008); Halliburton and KBR entities (\$579 million in 2009) and BAE Systems (\$400 million in 2010).

I consider an event-window that covers the period of proposal, debate and approval of the FCPA. That it, the event window goes from March 11, 1976 -introduction of the first bill regarding foreign bribery, which would be ultimately approved- to December 19, 1977 -when President Carter signed the FCPA into law-. I also extend the analysis to show the results one year before the proposal and one year after the approval, to control for trends. This allows me to show the stock market evolution after the first corruption scandals, that start in April 1975. These affected the

⁴This was one of the stiffest penalties ever written in the US Criminal Code. For example, it was twice as high as the maximum penalty for corporate price-fixing.

⁵The Alternative Fines Act allows the federal government to request twice the gain or loss associated with an offense if the Justice Department shows that a crime was committed.

firms involved,⁶ but it is not until March 1976 when a proposal of outlawing foreign payoffs was made. Actually, the official stance was to play down the scandals and, after June 1976, to require only disclosure. Having said that, I analyse the stock market performance for a period covering the first corruption scandals, the bill introduction and its debate and eventual approval.

2.3 Exposure to the FCPA

Companies that made use of foreign bribery, or potentially could make use of it, should be the most affected by the policy that prohibited questionable foreign payments. However, by their own nature, bribes are unobserved.⁷ As a consequence, I use indirect evidence. I take exports as a proxy for foreign operations. Firms more engaged in international trade conduct a higher amount of businesses abroad and therefore are more likely to engage in foreign payoffs.⁸

I gather data on NYSE firms from Compustat for the period 1970 to 1980.⁹ A crucial limitation of the dataset is that export data is available only since 1976, and only for a negligible subsample of firms¹⁰ This poses a significant problem, because exports in 1977 -the first year in which I have export data for a sufficient number of firms- could be the result of the policy itself, introduced in 1976. That is, the foreign bribery scandals in 1975, the debate of on the law in 1976-1977 and the evolution of the stock market in these years could affect the amount of exports, and therefore this simultaneity bias prevents from establishing a causal relationship regarding exports and stock market performance.

As a consequence, I resort to export data in 1974, the year before the first foreign bribery scandals hit the public opinion. With this I make sure that the proxy for exposure is not contaminated by the policy. As there are no data for firm exports in this year, I use aggregate US exports at the

⁶See Smith et al. (1984).

⁷In early 1975, following the first foreign bribery scandals, the SEC instituted a voluntary disclose programme in which corporations were encouraged to report voluntarily questionable foreign payments. Disclosure, although not exempting companies from prosecution, would reduce the likelihood of litigation against them. Under this programme, more than 500 corporations reported having resorted to this practice. It would be tempting to use these companies to construct a portfolio of exposed companies and compare it with a portfolio of non-reporting -non-exposed- companies. However, the fact that reporting was voluntary induces self-selection. Therefore, I would expect that the companies that reaped the highest benefits from foreign bribery would not choose to report, in order to not to jeopardize their network of corruptible officials. Moreover, the voluntary reporting extended from 1975 to the end of 1977, therefore, covering the whole period of debate and approval of the FCPA.

⁸An alternative is to use subsidiaries overseas as a proxy for foreign operations. Unfortunately, I do not have data on foreign direct investment (FDI) neither at the firm not at the sectoral level for the pre-policy period. Aggregate data show that trade flows were six times as high as FDI flows in the first half of the 1970s.

⁹See Appendix A for a list of the variables used throughout the paper, as well as definitions and sources.

¹⁰Just 12 firms of the final sample have export data in 1976. In 1977 the missing values are reduced, but still they account only for 58% of the final sample. Thus, it is not possible to gather *individual* export data *before* the policy was implemented.

four-digit SIC sectoral level, the highest disaggregation available in Feenstra (1997).¹¹¹²

The aim of this event-study is to build financial portfolios that differ in the exposure to international trade, and follow their returns during the event-window, that is, the period that goes from the introduction of the bill that proposes to outlaw foreign bribery until the bill is signed into law.

In order to calculate exposure, I define the Export Intensity of the 450 four-digit manufacturing SIC sectors in 1974 -the year prior to the first foreign bribery case- as:

Export Intensity_j = $\frac{\text{Exports}_j}{\text{Total Value of Shipments}_j}$

That is, the Export Intensity of sector j is the amount of Exports of that sector over the total output of the sector.¹³

Then, I rank all the sectors according to their Export Intensity and divide them in three reference groups each containing 150 sectors -low, medium and high exports-. Next, I assign each of the NYSE firms to a different group, according to the four-digit SIC of each company. For each group, I build a value-weighted buy-and-hold portfolio. In doing so, I restrict the analysis to manufacturing firms with complete data for the period 1970 to 1980. I also exclude the oil firms from the analysis, as these companies are most likely to be affected by the oil shocks of the 1970s. This leaves me with a sample of 356 manufacturing non-oil companies.¹⁴ Henceforth, I refer to these financial portfolios as the low, medium and high export portfolio, stemming from the assignment of NYSE companies to sector of low, medium and high exports, respectively.

¹¹Also, to explore the results when considering exports at the firm level, I perform a robustness check of the baseline results through an instrumental variables approach, performing an imputation to increase the sample size. See Subsection 4.2.

¹²Note that using sectoral export data is interesting by itself. Prohibiting foreign bribery may affect the stock price of firms engaged in international trade, as expected profits are lower. However, it may also lower the expected profits of firms not engaged in international trade but who operates in highly foreign exposed sectors, as the market would anticipate that these firms, in order to grow, would have to engage in foreign bribery in the future. Therefore, expected profits for these firms are also lower.

¹³Export intensity is very stable in the pre-policy period. For instance, the correlation between Export Intensity between 1973 and 1974 is .97 and between 1972 and 1974 is .96. An alternative to build the groups of low, medium and high export sectors is to classify them according to the mode of the groups each industry belongs to in the period 1970 to 1974. If I do this, results of Section 2.4 are even more pronounced. The high export portfolio experienced a negative cumulative abnormal return of 29.1 percent -instead of 26.1 percent- whereas the low export portfolio experienced a negative cumulative abnormal return of 0.1 percent -instead of 2.9 percent-. Still, I stick to the classification according to Export Intensity in 1974 given the stability of the variable and the proximity to the policy period.

¹⁴I construct a balanced panel in order to facilitate the building of the portfolios as well as the common risk factors of the Fama-French model as in Abadie and Gardeazabal (2003). I restrict to manufacturing firms -those with SIC between 2000 and 3999- as there are no data on industry exports outside manufacturing in Feenstra (1997). Oil firms are those belonging to SIC sectors between 1300 and 1399; between 2900 and 2999 and 3533 -23 companies of the balanced panel-.

Table I shows descriptive statistics of the sectors and firms that comprise each portfolio. The first three columns display the descriptive statistics of the low, medium and high export portfolio, respectively. The fourth column performs a test of the differences in means of the high minus the low export portfolio being different from zero.

Note that there are more sector and firms in the high export portfolio than in the low export portfolio. The average export intensity of the low export portfolio is 0.8 percent whereas it is 15.1 percent in the high export portfolio. Both sectors and firms in the high export portfolio are bigger in terms of value of shipments as well as market value, although only the latter difference is significantly different from zero. Firms in the high export portfolio also tend to have higher Tobin's Q and experienced a higher growth in real sales between 1973 and 1974.

Figure I shows the stock market evolution of the 3 portfolios during the whole period of interest. The event-window is between the red bars, corresponding to the 21 months that go from the introduction of the bill -March 11, 1976- to the FCPA signing into law -December 19, 1977-. Some important dates are also marked with grey bars. The graph is extended one year before the outset of the event-window and one year after its ending. The low, medium and high export portfolio are represented with dashed, dotted and solid lines, respectively. The cumulative returns are normalized to zero at the beginning of the event-window.

Note that the 3 portfolios perform in a remarkably similar fashion before the bill banning foreign bribery is introduced. From that moment on, they start to diverge. An investment in the value-weighted low export portfolio in March 11, 1976 would have yielded a 8.4 percent benefit in December 19, 1977. On the contrary, the same investment in the medium and high export portfolios would have yielded losses of 10.7 and 19.8 percent, respectively. Hence, during the event-window, the difference in returns between firms in sector of low export intensity versus firms in sectors of high exports intensity was as large as 28.2 percent. After the event-window, that is, once the policy is approved and the market has incorporated all the information, the 3 portfolios behave very similarly again.

2.4 Fama-French Model

One possible explanation for the discrepancy in the stock market returns of the 3 portfolios during the event-window may be related to different riskiness or characteristics of the portfolios. That is, literature has identified several equity characteristics that have explanatory power when analysing realized returns. If this characteristics differ systematically between the portfolios, this could explain at least some of the disparity in the observed returns.

Therefore, to analyse long-run stock returns it is necessary to control for this equity characteristics that help explain observed returns. The influential work by Fama and French (1993) identified three of these common risk factors, composing the so-called Fama-French Three-Factor Model. These three factors are exposure to the market return, size and book-to-market ratio. All these factors are related to fundamentals.¹⁵

¹⁵For instance, size is associated to profitability -small firms tend to have lower earnings on assets- whereas

	Low Exports	Medium Exports	High Exports	t test
	Mean (Std. Dev)	Mean (Std. Dev)	Mean (Std. Dev)	High - Low [p-value]
Panel A: Sectors				
Export Intensity	$0.0078 \\ (0.0054)$	$0.0399 \\ (0.0125)$	$\begin{array}{c} 0.1512 \\ (0.0895) \end{array}$	0.1434 [0.0000]
Total Value of Shipments	$2667 \\ (2351)$	$3349 \\ (5546)$	$3932 \\ (6121)$	$1265 \\ [0.1222]$
Cost of Energy	$0.0467 \\ (0.1044$	0.0407 (0.0509)	$0.0431 \\ (0.0619)$	-0.0036 $[0.8330]$
Number of Sectors	45	63	70	
Panel B: Firms				
Book to Market	2.6014 (3.0032)	$1.9899 \\ (1.3626)$	2.2764 (1.8499)	-0.3250 [0.4243]
Log Market Value	$\begin{array}{c} 4.3402 \\ (1.5710) \end{array}$	$4.9996 \\ (1.6943)$	$4.8194 \\ (1.7247)$	$0.4792 \\ [0.0497]$
Log Volume	$9.9512 \\ (1.5218)$	$10.0962 \\ (1.7464)$	$10.1209 \\ (1.6253)$	$0.1697 \\ [0.4661]$
Tobin's q	$0.8478 \\ (0.3883)$	1.0072 (0.6304)	$0.9594 \\ 0.5497$	$0.1116 \\ [0.0938]$
Dividend Yield	0.0847 (0.0590)	$0.0659 \\ (0.0371)$	$0.0725 \\ (0.0638)$	-0.0122 [0.1805]
Yearly Return	-0.2749 (0.2730)	-0.2357 (0.2533)	-0.2642 (0.3012)	$0.0108 \\ [0.7988]$
Real Sales Growth	-0.0191 (0.1581)	$0.0233 \\ (0.1015)$	$0.0208 \\ (0.1312)$	$0.0399 \\ [0.0795]$
Number of Firms	64	143	149	

TABLE I

Descriptive Statistics of Sectors and Firms Comprising Each Portfolio in 1974

Table I shows descriptive statistics of the sectors -Panel A- and firms -Panel B- that comprise the financial portfolios of low, medium and high export intensity. All the variables refer to the end of 1974, before the first foreign corruption scandals start. The fourth column reports the results of a test with null hypothesis being that the mean of the high export intensity portfolio is different from the mean of the low export intensity portfolio, allowing for unequal variances. See the Appendix for further information on the variables.

Hence, to account for this, I estimate for each portfolio the Fama-French Three-Factor Model with daily data for the period January 1970 to December 1974. I choose an estimation-window that does not overlap with the first corruption scandals (starting in April 1975), in order to guarantee that the estimates are not influenced by them.¹⁶ I estimate the following regression by OLS:

book-to-market is related to earnings -a high book-to-market implies sustained lower earnings-.

¹⁶An alternative is to estimate the Fama-French Three-Factor Model for the period January 1970 to March 1975. I choose not do so because in March 1975 the SEC moved against two major oil firms alleging that these companies kept slush funds for political contributions and other purposes, although it did not disclose who received the funds. Therefore, an estimation-window finishing in December 1974 ensures that the estimates are not contaminated by any foreign bribery issue. In any case, results do not change if I choose estimation windows finishing in March 1975



$$R_t^j = \alpha_j + \beta_1^j R_t^m + \beta_2^j SMB_t + \beta_3^j HML_t + AR_t^j \tag{1}$$

where R_t^j is the excess daily return -over the risk-free interest rate, the Fed Funds rate- of the buy-and-hold portfolio j in time t, R_t^m is the excess return on the market portfolio in time t, SMB_t ('Small Minus Big') is the difference between the returns of portfolios comprised of firms with small and big market capitalizations at time t, and HML ('High Minus Low') is the difference between the returns of portfolios composed by stocks with high and low book-to-market ratios at time t.¹⁷

Given the estimates $\hat{\alpha}_j$, $\hat{\beta}_1^j$, $\hat{\beta}_2^j$ and $\hat{\beta}_3^j$ I calculate daily out-of-sample abnormal returns as the part of the return not explained by the common risk factors (\hat{AR}_t^j) . Finally, in order to analyse the performance of the different portfolios, I customarily aggregate the abnormal returns of each portfolio calculating the cumulative abnormal return as:

$$\hat{CAR}_{t}^{j} = \left(\prod_{s=1}^{t} \{1 + \hat{AR}_{s}^{j}\}\right) - 1$$
(2)

Remember that the event-window goes from March 11, 1976 -day of the introduction of the bill- to December 19, 1977 -when the FCPA is signed into law-.

or March 1976 -just before the introduction of the foreign bribery bill-.

¹⁷See the subsection *Calculation of Portfolio Returns and Risk Factors* in Appendix B of Abadie and Gardeazabal (2003) for an explanation on how I compute the portfolio returns as well as the covariates of the Fama-French Three-Factor Model.

 $^{^{18}\}mathrm{Appendix}\;\mathrm{B}$ shows the results of the estimation.

Figure II shows the cumulative abnormal returns for the low, medium and high export portfolios during the period of interest. As before, the red bars delimit the event window, the trading days between the introduction of the bill to the signing into law. The cumulative abnormal returns are normalized to zero at the beginning of the event window.

Note that the cumulative abnormal returns of the high export portfolio start to fall precisely the day that the first legislative proposal regarding foreign bribery is made public. They keep falling during the event-window only to become flat after the FCPA is approved. On the contrary, the cumulative abnormal returns of the low export portfolio are near zero for the whole period of interest. For the medium export portfolio, the abnormal returns fall in between the low and high exports portfolio.

Before the outset of the event-window, there is no a clear pattern regarding the evolution of the abnormal returns of the portfolios. However, it is visually clear that the underperformance of firms in high exporting sectors during the debate and approval of the FCPA is not the consequence of a past trend.

Quantitatively, the high export portfolio experience a negative cumulative abnormal return of 26.1 percent, compared to also negative returns of 16.4 and 2.9 percent of the medium and low export portfolios, respectively. The difference between portfolios is in line with the differences in raw returns observed Figure I.



Figure II

CUMULATIVE ABNORMAL RETURNS OF THE 3 PORTFOLIOS

In the next section, I assess the statistical significance of these differences. Moreover, I control for industry shocks and more covariates that could explain stock market performance.

For the time being I explore two alternative explanations that could be behind the underper-

formance of firms in sectors most opened to international trade: the evolution of the exchange rate and the oil prices.

Figure III adds to Figure I the evolution of the nominal effective exchange rate. As is visible from the graph, the evolution of the dollar was very stable during most of the event window, suggesting the exchange rate is not behind the stock market returns of this period.¹⁹

FIGURE III

CUMULATIVE ABNORMAL RETURNS OF THE 3 PORTFOLIOS AND THE DOLLAR EXCHANGE



RATE

Another plausible explanation for the poor performance of firms in high export sectors during the event-window is the evolution of oil prices. Oil firms benefited from sharp increases in oil prices during the 1970s. For this reason, as stated above, I exclude them from the analysis. However, it remains the concern that sectors opened to international trade could be affected by higher costs of fuels, thereby explaining their underperformance in the stock market.

Figure IV plots the evolution of the log oil price on a monthly basis for the decade of the 1970s. The event-window is the period between the red bars.

Note that there are two oil shocks in this decade. The first one takes place in December 1974 and the second starts in April 1979. Oil prices increase by 184 percent and 166 percent in these shocks, respectively. In the event-window, oil prices increase by 23 percent. Therefore, the period of analysis is outside of the two major oil shocks that hit the global economy in this decade.

¹⁹Also, adding the exchange rate as an additional factor to the Fama-French model does not change the qualitative results. The daily return of the exchange rate only has explanatory power for the high export portfolio. Still, the cumulative abnormal returns of this portfolio if this augmented market model is estimated are -24.5 percent during the event-window.



Table II shows the correlation between the abnormal returns of the three portfolios and the daily change in oil prices between January 1977 and December 1980.²⁰. I find a negative correlation between the abnormal returns of the medium export portfolio and oil prices, significant at the 90 percent confidence level. For the low and high export portfolio, the coefficients are not significantly different from zero. Actually, the point estimate for the high export portfolio is positive. Overall, it seems that oil prices do not explain the abnormal returns of sectors most engage in international trade.

TABLE II
Abnormal Returns vs. Oil Prices.
JANUARY 3, 1977 TO DECEMBER 31, 1980

	,		,	
	Low Exp. (1)	Medium Exp. (2)	High Exp. (3)	
Oil Price Return	-0.0055 (0.0197)	-0.0326^{\sim} (0.0179)	0.0040 (0.0176)	
Constant	-0.0001 (0.0001)	-0.0004^{**} (0.0001)	-0.0004** (0.0001)	
Observations R-squared	$1,000 \\ 0.00$	$1,000 \\ 0.00$	$1,000 \\ 0.00$	

Table II shows the regression of the abnormal returns of each portfolio against the daily oil price return for the period January 3, 1977 to December 31, 1980. Robust standard errors in parenthesis. Significance levels: \sim : 10%; *: 5%; **: 1%.

 $^{^{20}1977}$ is the earliest year daily data on oil prices is available

3 Regression Results

The motivating evidence presented in the previous section suggested that firms in sectors more opened to international trade experienced negative stock market returns after the introduction of a bill that proposed the banning of foreign bribery. In this section, I explore further this result. In the next subsection I calculate the cumulative abnormal return of each firm in my sample during the event-window and regress it against the exposure to the policy and other covariates, which control for industry effects and other determinants of equity returns. In Subsection 3.2 I use the so-called Jensen's alpha approach in order to alternatively calculate the long-run abnormal returns of each portfolio during the event-window.

I explore further this result by means of a regression approach. This allows me to control for industry shocks and add other controls that could explain part of the observed abnormal returns.

3.1 Firms' CARs

In this subsection, I perform a regression of each firm cumulative abnormal return during the eventwindow against its exposure to the policy and other covariates, that control for other determinants of stock-market performance and industry shocks.

I proceed as follows. For each company in my sample (356 -the balanced panel-) I estimate the Fama-French Three-Factor Model for the period January 2, 1970 to December 31, 1974 -the same period as the previous analysis-. Then, I calculate the cumulative abnormal return of each firm for the event window -March 11, 1976 to December 19, 1977-, again as in the previous analysis.

Then, I estimate a cross sectional regression of the form:

$$\hat{CAR}_{i}^{j} = \beta_{1} + \beta_{2} Medium_{j} + \beta_{3} High_{j} + Industry Dummies + Controls_{i} + \epsilon_{i}$$
 (3)

where $C\hat{A}R_i^j$ is the cumulative abnormal return of firm *i* in sector *j* and Medium and High are dummies taking the value of 1 if sector *j* is a sector of medium or high exports, respectively;²¹. I also add industry dummies at the two-digit SIC level and several controls that can have predictive power in explaining out-of-sample abnormal returns. These controls are taken as of December 31, 1974, that is, before the foreign corruption scandals are made public, to guarantee that they are not contaminated by them. Observations are weighted by market value as of December 31, 1974 ²² and standard errors are clustered at the four-digit SIC level.

Table III displays the results of estimating equation (3). In column (1) we see that firms in sectors of high exports experienced an additional 24.1 percent drop on average in cumulative

 $^{^{21}\}mathrm{The}$ excluded dummy, a sector of low exports, is captured by the constant when I exclude the two-digit industry dummies

 $^{^{22}}$ I weight observations for two reasons. First, by weighting you obtain a more accurate measure of the benefit or loss that a potential investor would incur if she takes a financial position in the period of interest. And second, asset pricing models have more troubles to explain the returns of small firms, which are more illiquid. See Fama (1998) for a discussion on weighting in calculating long-run abnormal returns.

abnormal returns -compared to firms in sectors of low exports- during the period that goes from the introduction of the bill banning foreign bribery to the moment it becomes law. This result -without controls- is the regression equivalent to Figure II.²³

In column (2) I add two-digit industry dummies in order to account for industry shocks. Note that identification is given by the fact that the industry dummies are at the two-digit SIC level whereas the variable of export intensity is at the four-digit SIC level. That is, I exploit the extra variation of sector Export Intensity within two-digit SIC industries to guarantee identification. Therefore, I can account for broad industry shocks in analysing the abnormal returns of firms that operate in sectors that differ in their export intensity. I argue that industries defined at the two-digit SIC level are a sufficiently narrow definition of sectors in order to account for industry shocks. Note that the SIC method of classifying industries goes from general to particular industries according to the production process. Then, they are very suitable to account for technology shocks, even if not the highest disaggregation is taken into account.

The results show that within industries, firms in sectors most engaged in international trade experienced a significant 25.4 percent drop in stock market returns.

In column (3), I add several covariates that could explain the observed realized returns. Although the Fama-French model already controls for factors such as size and book-to-market, it is useful to control for other financial variables that can explain realized returns, as the dependent variable is computed out-of-sample. Then, I control for size, Tobin's Q, volume, the price level, dividend yield, the return on the stock one year before and the previous growth rate in real sales. All the controls are taken as of December 31, 1974, to guarantee that they are not contaminated by events regarding foreign bribery. I also include the cost of energy at the four-digit SIC level to control for input expenditures that can be affected by oil prices.

The previous results are robust when I control for all these covariates. The high export portfolio experiences a significant negative abnormal return with respect to the low export portfolio. The medium export portfolio also experiences a negative return, but less than half the size of the one of the high export portfolio. Interestingly, I find a significant negative effect of the cost of energy, meaning that firms in sectors with higher expenditures in electric and fuels underperformed in the stock market, perhaps because the stock market anticipated future increases in energy prices.

Certainly, the classification of NYSE companies in three portfolios is arbitrary. However, it guarantees a suitable amount of firms -and sectors- to build the portfolios and measure precisely their stock market evolution. In Table IV I consider an alternative number of portfolios in classifying sectors by their Export Intensity. They are all constructed as having the same number of four-digit SIC sectors and according to the Export Intensity in 1974 -before the foreign corruption

²³The cumulative abnormal returns of the three portfolios, however, are not numerically the same. This is because the cumulative of weighted averages is different from the weighted average of the cumulatives. The residuals of the Fama-French model when estimated for the portfolios are a weighted average of the residuals stemming from fitting the Fama-French model for every individual firm belonging to each portfolio. However, Figure II displays the cumulative of the out-of-sample residuals whereas results in Table III shows a weighted average of the cumulative abnormal returns of the firms.

	(1)	(2)	(3)
Medium Export Intensity	-0.1011 (0.0653)	-0.0096 (0.0729)	-0.0845^{\sim} (0.0464)
High Export Intensity	-0.2415^{**} (0.0555)	-0.2544^{**} (0.0912)	-0.2226^{**} (0.0565)
Constant	-0.0695 (0.0447)		
Cost of Energy			-1.1757^{**} (0.2292)
Log Market Value			-0.081^{**} (0.0200)
Log Volume			0.0699^{**} (0.0126)
Tobin's Q			-0.0229 (0.0192)
Price Level			-0.0014 (0.0013)
Dividend Yield			1.7663^{*} (0.6870)
Last Year Return			-0.1564^{\sim} (0.0932)
Last Year Real Sales Growth			-0.2357 (0.1667)
Industry Dummies	NO	YES	YES
Observations	356	356	356
R-squared	0.08	0.23	0.49

TABLE III Regression Analysis: 3 Portfolios of US Firms

Table III shows the regression of the firm cumulative abnormal return for the period March 11, 1976 to December 19, 1977 against the export intensity of the firm sector and several controls. That is, the table shows the results of equation 3 in text. Observations are weighted by market value as on December 31, 1974. Control variables are taken as on December 31, 1974. Industry dummies correspond to two-digit SIC manufacturing sectors. Robust standard errors are in parenthesis, clustered at the four-digit SIC sector. Significance levels: \sim : 10%; *: 5%; **: 1%. See the Appendix and the text for further details.

scandals-. In columns (1) and (2) I consider a split in six portfolios and in columns (3) and (4) a split in ten portfolios. Note that the covariates referring to Portfolios in Table IV are not the same between specifications (1)-(2) and (3)-(4). For instance, Portfolio 6 in (1)-(2) refers to the portfolio of highest export intensity whereas in (3)-(4) it refers to the portfolio 5th in ranking of export intensity -Portfolio 10 is the first in the ranking-. The controls and industry dummies included are the same as in Table III

The results under these alternative classifications prevail. Firms in sectors most engaged in international trade significantly underperform sectors of lower exports. Actually, the non-linearity of this results seems apparent. Firms in sectors above the median of Export Intensity -Portfolios 4 to 6 in columns (1) and (2) and Portfolios 6 to 10 in columns (3) and (4)- experience negative abnormal returns during the event window. On the other hand, sectors below the median of Export Intensity experience near zero abnormal returns

Finally, columns (5) and (6) of Table III show the results of estimating a pure linear model of the effect of Export Intensity on cumulative abnormal returns. They display the coefficient estimates of the following regression:

$$\hat{CAR}_{i}^{j} = \alpha_{1} + \alpha_{2}X_{j} + \text{Industry Dummies} + \text{Controls}_{i} + \xi_{i}$$
 (4)

where X_j is the Export Intensity of sector j and the rest of variables are defined as before.

I find a negative coefficient on the Export Intensity variable, ranging from -0.71 to -0.50 depending on the covariates considered. This means -taking the coefficient in the full specification with controls- that an increase of 10 percent in the export intensity is associated with a 5 percent decrease in the stock market cumulative abnormal return during the event-window. Precision in the estimates, though, is lower, specially when including controls.

3.2 Jensen's Alpha Approach

To further assess the abnormal performance of the portfolios considered, I explore how the portfolios behave under a model of performance attribution -the so-called Jensen's alpha approach or the calendar-time portfolio approach. This is a method widely used in the finance literature to estimate risk-adjusted abnormal performance (see for example Loughran and Ritter (1995), Brav and Gompers (1997), Dichev and Piotroski (2001) and Gompers et al. (2003)).

The Jensen's alpha approach has several characteristics that make it similar to the one presented above, although with differences. Previously, I used the elasticities of the portfolios with respect to the common risk factors estimated in the pre-policy period to make an expectation of the *normal* performance during the event-window. Therefore, I interpreted the difference between the actual and the *expected* returns -the out-of-sample residuals- during the event-window as the abnormal returns.

The Jensen's alpha approach, on the contrary, estimates the market model only for the eventwindow.²⁴ The underlying rationale is that the common risk factors should explain the performance of the portfolio. Therefore, it can be interpreted as a performance attribution method. The estimated intercept -the alpha- can then be thought of as the abnormal return in excess of what would have been achieved by investing in the common risk factors. That is, if the different performance is a manifestation of a diverse sensitivity with respect to the common risk factors, then the estimated alpha should be statistically not different from zero.

Note that with this approach I lose the graphical analysis presented in Figure II, which was notorious in accounting for the fact that the cumulative abnormal returns start to diverge between

 $^{^{24}\}ensuremath{\mathrm{Therefore}}$, by construction the estimated residuals sum up to zero.

	6 Por	tfolios	10 Por	10 Portfolios		Linear	
	(1)	(2)	(3)	(4)	(5)	(6)	
Portfolio 2	-0.0841 (0.0877)	-0.0781 (0.0862)	-0.0446 (0.0736)	-0.0085 (0.1262)			
Portfolio 3	-0.0471 (0.0810)	-0.0672 (0.0742)	-0.0962 (0.1289)	-0.025 (0.1081)			
Portfolio 4	-0.2514^{**} (0.0748)	-0.2858^{**} (0.0890)	$0.0387 \\ (0.1639)$	$0.0759 \\ (0.1737)$			
Portfolio 5	-0.2789^{**} (0.0784)	-0.2597^{**} (0.0823)	-0.0943 (0.0731)	-0.0984 (0.0925)			
Portfolio 6	-0.2694^{**} (0.0536)	-0.2841^{**} (0.0834)	-0.3103^{**} (0.0837)	-0.1792 (0.1226)			
Portfolio 7			-0.1376^{\sim} (0.0784)	-0.269^{*} (0.1266)			
Portfolio 8			-0.2603^{**} (0.0788)	-0.2679^{*} (0.1168)			
Portfolio 9			-0.3367^{**} (0.0788)	-0.2284^{*} (0.1026)			
Portfolio 10			-0.2256^{**} (0.0790)	-0.2728^{*} (0.1196)			
Export Intensity					-0.7065^{*} (0.3380)	-0.5007^{\sim} (0.2821)	
Constant	-0.0341 (0.0454)		-0.0288 (0.0582)		-0.168^{**} (0.0428)		
Industry Dummies Controls	NO NO	YES YES	NO NO	YES YES	NO NO	YES YES	
Observations R-squared	$\begin{array}{c} 356 \\ 0.13 \end{array}$	$\begin{array}{c} 356 \\ 0.52 \end{array}$	$356 \\ 0.19$	$\begin{array}{c} 356 \\ 0.53 \end{array}$	$\begin{array}{c} 356 \\ 0.04 \end{array}$	$\begin{array}{c} 356 \\ 0.47 \end{array}$	

TABLE IV Regression Analysis: Additional Number of Portfolios of US Firms

Table IV shows the regression of the firm cumulative abnormal return for the period March 11, 1976 to December 19, 1977 against the export intensity of the firm sector and several controls. Columns (1) & (2) consider the regression against six portfolios of export intensity, whereas columns (3) & (4) consider 10 portfolios of export intensity. Note, therefore, that the covariates referring to Portfolios are different between specifications (1)-(2) and (3)-(4). For instance, Portfolio 6 in (1)-(2) refers to the portfolio of highest export intensity whereas in (3)-(4) it refers to the portfolio 5th in ranking of export intensity -Portfolio 10 is the 1st in the ranking-. The last two columns regress the dependent variable against the export intensity of the sector, under a linear specification. Controls and industry dummies are the same as in Table III. Observations are weighted by market value as on December 31, 1974. Control variables are taken from 31 December, 1974. Industry dummies correspond to two-digit SIC manufacturing sectors. Robust standard errors are in parenthesis, clustered at the four-digit SIC sector. Significance levels: ~: 10%; *: 5%; **: 1%. See the Appendix and the text for further details.

portfolios when the FCPA was introduced in the political debate. However, it has the advantage that it estimates the elasticities with respect to the common risk factors in the event-window, and therefore excludes the possibility that structural changes in these elasticities aside from the actual policy considered could partly explain the evolution of the abnormal returns. Panel A of Table V shows the results of estimating Equation (1) with daily data for each of the three portfolios with the event-window as the sample period. The estimated alpha of the high exports portfolio is a significantly negative 0.03 percent per day, which yields a negative abnormal return of 12.2 percent during the event-window. The medium export intensive portfolio also experiences a negative abnormal return of around 11.0 percent during the event-window. On the contrary, the alpha of the low exports portfolio is not statistically different from zero, suggesting no abnormal returns during the event-window.

Panel B changes the regressions' sample period to explore the performance of the portfolios during the period that goes from the first foreign corruption scandal to the day before the beginning of the event-window. For all three portfolios, the estimated intercepts are not statistically different from zero.

Finally in Panel C I run the same regressions but for one year after the approval of the FCPA. Again, none of the alphas are estimated to be different from zero. This suggests that the market discounted the effect of the policy once it was approved and therefore, the performance of the portfolios from that moment on is similar.

4 Further Evidence

In this section I perform two robustness checks of the previous findings. First, I perform the analysis for a country not subject to the FCPA -the United Kingdom- in order to disregard the existence of global shocks that could drive the results. Second, I make use of firm-level export data through an instrumental variables approach and test whether the results prevail.

4.1 Results for United Kingdom

Only firms listed on the US were subject to the Foreign Corrupt Practices Act. Therefore, companies outside the US should benefit or at least not be affected by this policy. Also, analysing the returns of companies outside the US is useful in which it allows to perform a robustness check on the existence of global shocks to trade that could explain the observed stock market returns. In this sense, I look at the returns of firms in the United Kingdom (UK) during this period.²⁵.

I collect data for UK firms for the period 1970 to 1980 from Datastream. One limitation of the dataset is that the industry classification of Datastream -Industry Classification Benchmark (ICB)is different from the one provided by Compustat for US firms -Standard Industrial Classification (SIC)-.

To solve this problem, I do a mapping between the two, based on industry definitions. Unfortunately, information is lost. There are much more industry categories in the four-digit SIC -1009-

²⁵The reason to choose the United Kingdom as a comparison is twofold. First, it is an economy very closed to the US, therefore, global shocks are most likely to affect both of them. And second, it is the only country for which I can gather data on firm stock returns for the whole decade 1970 to 1980. Hence, I can perform the closest analysis I carried out in section 2

	Alpha	R^m	SMB	HML	Obs.	R-squared
Panel A: Event-Window						
Low Export Intensity	-0.0103 (0.0119)	87.6012^{**} (2.0898)	$24.4445^{**} \\ (4.6279)$	18.0682^{**} (4.2745)	446	0.81
Medium Export Intensity	-0.0260^{**} (0.0089)	106.1564^{**} (1.5552)	-0.1699 (3.4440)	-9.2241^{**} (3.1810)	446	0.92
High Export Intensity	-0.0292^{**} (0.0100)	$\frac{118.5373^{**}}{(1.7507)}$	-24.3265^{**} (3.8770)	-25.3001^{**} (3.5809)	446	0.93
Panel B: First Corruption	Scandal - Be	ginning of Ev	ent-Window	(Apr-75 to Ma	ur-76)	
Low Export Intensity	0.0064 (0.0239)	94.4832^{**} (2.7335)	31.6093^{**} (6.1657)	2.064 (5.4910)	233	0.85
Medium Export Intensity	$0.0046 \\ (0.0181)$	$\begin{array}{c} 112.1096^{**} \\ (2.0759) \end{array}$	-5.1696 (4.6824)	-9.4073^{*} (4.1700)	233	0.93
High Export Intensity	$0.0204 \\ (0.0169)$	117.4225^{**} (1.9348)	-11.0122^{*} (4.3641)	-11.6964^{**} (3.8865)	233	0.95
Panel C: One Year after F	CPA (Dec-77	' to Dec-78)				
Low Export Intensity	-0.0129 (0.0173)	92.464^{**} (2.3557)	36.2453^{**} (4.2546)	-29.4409^{**} (6.5874)	253	0.88
Medium Export Intensity	-0.0043 (0.0115)	108.614^{**} (1.5597)	-9.3422^{**} (2.8170)	-19.5583^{**} (4.3616)	253	0.96
High Export Intensity	$0.0190 \\ (0.0123)$	124.4016^{**} (1.6796)	-18.6286^{**} (3.0335)	-25.999^{**} (4.6967)	253	0.96

		Table V	Τ	
Jensen's	Alpha	Approach	FOR 3	Portfolios

Table V shows the regressions of the daily returns of the Low, Medium and High Export Intensity Portfolios against the common risk factors of the Fama-French Model. The constant of these regressions can be interpreted as a measure of the abnormal performance of the Portfolio (Jensen's Alpha approach). Panel A shows the regression for the event-window (Mar-76 to Dec-77). Panel B for a period before the event-window, from the first corruption scandal -April 8, 1975- to the day before the beginning of the event-window -March 10, 1976-. Panel C shows the results of the three regressions for a period extending one year after the FCPA was signed into law -December 20, 1977 to December 20,1978-. Estimated coefficients as well as standard errors are multiplied by 100 for presentational purposes. Standard errors are in parenthesis. Significance levels: \sim : 10%; *: 5%; **: 1%. See the Appendix and the text for further details.

than in the Subsectors ICB -114-. Eventually, I map 369 manufacturing SIC sectors -out of 450-into 40 ICB sectors.²⁶

Then, I proceed as before. I divide the 40 manufacturing ICB sectors in three groups according to the US Export Intensity in 1974 and then assign US and UK companies to one group according to their industry. Then, for each group and each country, I build a value-weighted buy-and-hold portfolio. I look at abnormal returns by estimating a Market Model²⁷ for the period 1970-1974 and then compute out-of-sample abnormal returns. I calculate then the cumulative abnormal returns

²⁶This mapping is available upon request.

 $^{^{27}}$ Due to data limitations regarding UK firms the Market Model only contains as covariates the Market Return

and follow them during the event window -March 11, 1976 to December 19, 1977- and one year before and after, to control for trends.

Figure V shows the evolution of the stock market returns of these portfolios for the US -Panel A- and the UK -Panel B-. Although the classification of firms in rougher due to the smaller number of industries, Panel A of Figure V show that US firms in ICB sectors more opened to international trade experienced a negative return during the event window whereas firms in ICB industries of low exports experienced a near zero return -compare it with Figure I, where four-digit SIC industries were used-.

Panel B of Figure V shows that in United Kingdom there is no difference in cumulative returns between firms in sectors in which the US had different exposure to international trade. The three portfolios perform in a remarkably similar fashion during the whole period of interest.

Figure VI shows for both countries the evolution of the abnormal returns. For the US -Panel A- I find that the abnormal returns are significantly negative for companies in ICB sectors more open to international trade. On the contrary, as we can see in Panel B, in the United Kingdom, the evolution of the cumulative abnormal returns of the portfolios is very noisy and indistinguishable.

Then, it seems that the pattern we observe for the US regarding companies in sectors with different exposure to international trade does not show up in the United Kingdom. Then, I can rule out broad industry shocks or global shocks to trade that can explain the patterns observed.

To compare the evolution of the portfolios for United States and United Kingdom in a more systematic way, I perform a differences-in-differences regression, adding some controls to check for robustness. I run the following regression:

$$C\widehat{A}R_{i}^{jc} = \beta_{0} + \beta_{1}US + \beta_{2}Medium_{j} + \beta_{3}High_{j} + \beta_{5}US * Medium_{j} + \beta_{6}US * High_{j} + Industry Dummies + Controls + \epsilon_{i}$$
(5)

where $C\hat{A}R_i^{jc}$ is the cumulative abnormal return²⁸ during the event window of firm *i* in sector *j* in country *c*; US is a dummy taking value 1 if *c* equals the United States and Medium and High are dummies taking the value of 1 if ICB industry *j* is a US medium or a high export sector. I also add Industry Dummies at the level of ICB Supersectors. Controls include the Log of Market Value and the Cost of Energy, taking as of December 31, 1974. Observations are weighted by market value as of December 31, 1974. The weight of a firm is relative to the rest of companies in its country.²⁹. Standard errors are clustered at the level of ICB Supersectors.

Columns (1) to (3) of Table VI shows the results. Only the interactions of the US dummy and the Medium and High Export dummies are shown. As we can see, in all specifications firms in high export sectors experience a negative abnormal return compared to firms in the same industries in the United Kingdom. This underperformance is 29.2 percent in the regression with industry dummies and controls.

 $^{^{28}}$ Computed as in Section 2.4 with the difference that the market model only controls for the market return due to data limitations. The market return for UK firms is the FTSE All Shares Index, provided by Datastream

 $^{^{29}\}mathrm{I}$ avoid overweighting of US firms as they tend to be bigger than UK companies.

FIGURE V





In columns (4) to (6) I perform the same differences-in-differences regression but allowing for a linear relationship. That is, I estimate the following regression:

$$\hat{CAR}_{i}^{jc} = \alpha_{1} + \alpha_{2}X_{j} + \alpha_{3}X_{j} * \text{US} + \text{Industry Dummies} + \text{Controls} + \xi_{i}$$
(6)

where X_j is the Export Intensity of ICB sector j.

The results show that the additional effect -relative to the United Kingdom- of being a US

FIGURE VI

US & UK CUMULATIVE ABNORMAL RETURNS OF PORTFOLIOS DEFINED FROM ICB INDUSTRIES



company in a sector with a higher Export Intensity is significantly negative.

4.2 Firms' Exports Data

Compustat provides data on exports at the firm level starting in 1976. However, for this year, only data for 12 firms -out of 356- are available. For 1977, the number of missing values is still large -58

		3 Portfolios				
	(1)	(2)	(3)	(4)	(5)	(6)
Medium Exports * US	-0.0352 (0.0863)	-0.0479 (0.1454)	-0.1552 (0.1371)			
High Exports $*$ US	-0.2613^{**} (0.0908)	-0.3563^{*} (0.1439)	-0.2920^{*} (0.1170)			
US	0.0527 (0.0633)			$0.0116 \\ (0.0743)$		
Export Intensity * US				-1.3784 (0.9281)	-2.3733^{*} (0.9941)	-1.5370^{\sim} (0.8867)
Industry Dummies Controls	NO NO	YES NO	YES YES	NO NO	YES NO	YES YES
Observations R-squared	$\begin{array}{c} 490\\ 0.11 \end{array}$	$490 \\ 0.27$	$\begin{array}{c} 490\\ 0.43\end{array}$	$\begin{array}{c} 490\\ 0.06\end{array}$	$\begin{array}{c} 490\\ 0.24\end{array}$	$\begin{array}{c} 490\\ 0.43\end{array}$

TABLE VI US vs. UK: Differences in Differences

Table VI shows the differences-in-differences regression of the US & UK firms' cumulative abnormal returns. Columns (1) to (3) show the results of the differences-in-differences regression of the 3 benchmark portfolios -equation (5) whereas columns (4) to (6) shows the results on the differences-in-differences of the linear specification -equation (6)-. Only the coefficients of the interactions of the export intensity and the US dummy are shown. Controls -and their interactions with the US dummy- include Log of Market Value and Cost of Energy. Observations are weighted by market value as on December 31, 1974. A firm weight is relative to the size of the firms in its country. Control variables are taken from 31 December, 1974. Industry dummies correspond to ICB Supersectors. Robust standard errors in parenthesis, clustered at the ICB sector. Significance levels: $\sim: 10\%$; *: 5%; **: 1%. See the text for further details.

percent of the sample-. Apart from the small sample size, data on firm exports in 1977 poses an additional significant problem. The amount of foreign sales in 1977 can be affected by the foreign corruption scandals of 1975 or the anticipation of a law approval in 1976 and 1977. Moreover, they could be the result of the evolution of the market value of firms during these months. In other words, there could be a simultaneity bias in the relation between firm exports in 1977 and stock market performance in 1976-1977.

For these reasons, in the main body of results I have resorted to industry export data to account for exposure to the US anti-bribery legislation. However, as a robustness check, I test whether results prevail when using the individual export data. I address the two problems with these data in the following way.

First, to account for the simultaneity bias, I adopt an instrumental variables approach. I instrument the exports of the firm in 1977 with sectoral exports in 1974. In the first stage I show that the sector of the firm in 1974 is a significant predictor of the firm exports in 1977. As there are no foreign corruption scandals before 1975, sectoral exports in 1974 are not contaminated by foreign bribery in any respect.

Second, in order to increase the sample size, I use the coefficients of the first stage regression to impute the value of exports in 1977 -net of the rest of the exogenous covariates-. This allows me to perform the 2SLS regression for the 356 companies of my sample.

Then, I start by defining Firm Export Intensity of firm i in 1977 as:

$$\label{eq:Firm_Export_Intensity} \text{Firm Export Intensity}_i = \frac{\text{Export sales}_i}{\text{Total Sales}_i}$$

That is, I calculate the Export Intensity of a company as the share of total sales that go to foreign countries.

Next, I run the following regression by 2SLS:

$$C\hat{A}R_i^j = \alpha_1 + \alpha_2 \text{Firm Export Intensity}_i + \text{Industry Dummies} + \text{Controls}_i + \xi_i$$
 (7)

where I instrument Firm Export Intensity_i with the (Sectoral) Export Intensity_i in 1974.

Table VII shows the results of the first stage regression. In all specifications, with and without industry dummies and controls, the Export Intensity of the firm sector in 1974 is a significant predictor of the firm's exports in 1977.

FIRMS EXPORTS: FIRST STAGE REGRESSION							
	(1)	(2)	(3)				
Sector Export Intensity	0.5639^{*} (0.2534)	0.4821^{*} (0.1893)	0.5553^{**} (0.1890)				
Constant	0.0143^{\sim} (0.0232)						
Industry Dummies Controls	NO NO	YES NO	YES YES				
Observations R-squared	$\begin{array}{c} 150 \\ 0.26 \end{array}$	$\begin{array}{c} 150 \\ 0.57 \end{array}$	$\begin{array}{c} 150 \\ 0.67 \end{array}$				

TABLE VII

Table VII shows the regression of the firms relative exports in 1977 against the Export Intensity of the Sector in 1974 and several controls. These include Cost of Energy, Log Market Value, Log Volume, Tobin's Q, Price Level, Dividend Yield, Last Year Return, Last Year Real Sales Growth. Observations are weighted by market value as on December 31, 1974. Industry dummies correspond to two-digit SIC sectors Robust standard errors in parenthesis, clustered at the four-digit SIC sector. Significance levels: \sim : 10%; *: 5%; **: 1%. See the text for further details.

The results of the second-stage are shown in Table VIII. The first three columns show the 2SLS regression of the firm cumulative abnormal return during the event window against the Firm Export Intensity in 1977 and several controls. The Firm Export Intensity in 1977 is instrumented with the sectoral Export Intensity in 1974 -before the first foreign corruption scandals-.

As we can see in columns (1) to (3), I find that a higher Export Intensity of the firm is associated with a lower cumulative abnormal return. The magnitude of these coefficients are lower than the ones I found when controlling directly for the sectoral Export Intensity -see columns (1) to (3) in

Table III-. Precision, however, is lower. The effect is statistically different from zero with a 90 percent confidence level when I add the full set of industry dummies and controls.

In columns (4) to (6) of Table VIII I explore the existence of a non-linear effect. I classify the fitted and imputed values of the first stage regression in three groups, each containing the same number of firms.³⁰ Then I define three dummy variables that identifies each group -Low, Medium and High Export Firms- and use them as covariates for the second stage regression.

I find that firms with Medium and High Exports perform worse in the event-window although this difference is not statistically different from zero without controls. When I add a set of covariates that can explain abnormal returns, I find that firms that had High Exports in 1977 -instrumentedexperienced an negative abnormal return of 17.1 percent on average during the event window compared to a group of firms with Low Exports.

Overall, despite the limitations of the data, the results that firms more engaged in international trade experienced negative abnormal returns during the debate and approval of the FCPA seem to prevail.

5 Conclusions

I provide evidence of a striking stock market underperfomance of firms in industries engaged in international trade during the period March 1976 to December 1977. I find that these companies experienced a significant negative cumulative abnormal return of around 22 to 25 percent during this time.

This period coincides with the debate and approval of a policy in the US that unilaterally prohibited banning foreign officials when conducting international business transactions -the FCPA-. I argue, therefore, that a suitable candidate for the smoking gun is precisely this policy. The market would have interpreted that the unilateral prohibition put US companies in disadvantage with respect to foreign competition.

Since the 1970s a large stream of foreign corruption scandals have shown that foreign bribery is prevalent and widespread when firms engage in international business transactions. The unilateral banning of this practice in the US in 1977 offers an attractive experiment to study the value of these payoffs and how firms were affected by the prohibition.

I conduct a long-run event-study analysis to investigate the stock market performance of firms

³⁰This is a different way of proceeding with respect to the results of columns (1) to (3) of Table III in Section 3, where I classified the universe of manufacturing sectors in the US in three equal-sized groups, and then assigned NYSE firms to each group. For this reason, the benchmark portfolios of low, medium and high export did not contain the same number of companies. In current analysis, I classify the fitted values of the first-stage regression in three equal-sized groups. Therefore, the coefficients of columns (4) to (6) of Table VIII are not strictly comparable with those of columns (1) to (3) in Table III, although both group of coefficients take 3 groups as a benchmark. Actually, we would expect the coefficients of Table VIII to be higher, as the measure of exposure -the 3 groups- is relative to just the sample of firms. On the contrary, in Table III exposure is with respect to the whole manufacturing sector in the US economy.

	Linear				3 Portfolios	3
	(1)	(2)	(3)	(4)	(5)	(6)
Firm Export Intensity	-1.2529^{*} (0.5994)	-1.7056 (1.1233)	-0.9016^{\sim} (0.5080)			
Firm Medium Exports				-0.0334 (0.0610)	-0.0202 (0.0562)	-0.0582 (0.0503)
Firm High Exports				-0.0481 (0.0479)	-0.0821 (0.0591)	-0.1708^{**} (0.0541)
Constant	-0.1501^{**} (0.0495)			-0.1930^{**} (0.0335)		
Cost of Energy			-1.4976^{**} (0.2401)			-1.4448^{**} (0.1968)
Log Market Value			-0.0681^{**} (0.0217)			-0.0727^{**} (0.0184)
Log Volume			0.0724^{**} (0.0132)			0.0742^{**} (0.0135)
Tobin's Q			-0.0567^{*} (0.0226)			-0.0621^{*} (0.0248)
Price Level			-0.0007 (0.0012)			-0.0006 (0.0012)
Dividend Yield			1.4553^{*} (0.7110)			1.3920^{*} (0.7003)
Last Year Return			-0.2008^{*} (0.1011)			-0.2390^{*} (0.1007)
Last Year Real Sales Growth			-0.2500 (0.1919)			-0.2905 (0.1825)
Industry Dummies	NO	YES	YES	NO	YES	YES
Observations	356	356	356	356	356	356

Table VIII

CUMULATIVE ABNORMAL RETURNS VS. FIRM EXPORTS (SECOND STAGE)

Table VIII shows the regression of the firm cumulative abnormal return for the period March 11, 1976 to December 19, 1977 against the Firm Export Intensity in 1977 instrumented with the Sector Export Intensity in 1974 and several controls. Columns (1) to (3) specify a linear relationship between the abnormal returns and the Firm Export Intensity. Columns (3) to (6) classify the fitted values of the first stage regression in 3 groups and then regress the cumulative abnormal returns -second stage- against dummies of the groups. I use the first stage regression to impute the Firm Export Intensity for companies that do not have data on Firm Export Intensity in 1977. Observations are weighted by market value as on December 31, 1974. Control variables are taken from 31 December, 1974. Industry dummies correspond to two-digit SIC manufacturing sectors. Robust standard errors in parenthesis, clustered at the four-digit SIC sector. Significance levels: $\sim: 10\%$; *: 5%; **: 1%. See the text for further details.

differing in exposure to this policy. As bribes are unobserved, I use exports as a proxy for foreign operations and therefore as a proxy for the likelihood of resorting to questionable payments.

The analysis, however, is difficult in its own nature. The underlying assumption is that during the event-window -March 1976 to December 1977- no other shocks affected systematically firms in heterogeneous sectors regarding foreign trade. The fact that the event-window spans 21 months and that the 1970s is a turbulent decade in the world economy poses some concerns regarding this assumption.

Moreover, the whole analysis relies on indirect evidence. The fact that bribes both in the intensive and in the extensive margin are unobserved establish a significant limitation. This is a common drawback that papers in the forensic economics literature share. Nevertheless, the issue of foreign bribery is intrinsically linked to the fact of conducting businesses abroad in general and to exports in particular.

Still, the FCPA behaves well as a candidate for explaining the underpeformance of firms opened to international trade during the event-window. I show that the cumulative abnormal returns of these companies start to fall precisely the day the FCPA is introduced in the political debate and keep falling until it is signed into law. I also show that the negative stock market performance is also evident when considering simply raw returns and when using exports at the level of the firm. Moreover, it is also robust to the inclusion of several covariates that could have explanatory power in out-of-sample abnormal returns. Also, the oil shocks of the 1970s are out of the period of interest. Finally, the underpeformance is not present when I apply the same analysis to the United Kingdom, a country that was not affected by the US policy.

Given this evidence, I conclude that the stock market evolution during the event window seems to be explained -at least partly- by the issue of foreign bribery. The fact that US corporations were prohibited from resorting to this practice whereas foreign competition was not seems to have decreased expected profits of the companies most affected.

References

- Abadie, Alberto and Javier Gardeazabal, "The Economic Cost of Conflict: A Case Study of the Basque Country," *American Economic Review*, March 2003, 93 (1), 113–132.
- Banerjee, Abhijit, Rema Hanna, and Sendhil Mullainathan, "Corruption," February 2009. Working Paper.
- Bartelsman, Eric J. and Wayne Gray, "The NBER Manufacturing Productivity Database," October 1996. NBER Technical Working Paper No. 205.
- Beck, Paul J., Michael W. Mather, and Adrian E. Tschoegl, "The Impact of the Foreign Corrupt Practices Act on US Exports," *Managerial and Decision Economics*, 1991, 12, 295–303.
- Brav, Alon and Paul A. Gompers, "Myth or Reality? The Long-Run Underperformance of Initial Public Offerings: Evidence from Venture and Nonventure Capital-Backed Companies.," *The Journal of Finance*, December 1997, 52 (5), 1791–1821.
- Campbell, John Y., Andrew W. Lo, and Craig A. MacKinlay, The Econometrics of Financial Markets, Princeton, NJ: Princeton University Press, 1997.
- **DellaVigna, Stefano and Eliana La Ferrara**, "Detecting Illegal Arms Trade," December 2009. Working Paper.
- Dichev, Ilia D. and Joseph D. Piotroski, "The Long-Run Stock Returns Following Bond Rating Changes," *The Journal of Finance*, February 2001, 56 (1), 173–203.
- **D'Souza, Anna**, "The OECD Anti-Bribery Convention: Changing the Currents of Trade," March 2009. Working Paper.
- Dube, Arindrajit, Ethan Kaplan, and Suresh Naidu, "Coups, Corporations, and Classified Information," March 2008. Working Paper.
- Faccio, Mara, "Politically Connected Firms," American Economic Review, March 2006, 96 (1), 369–386.
- Fama, Eugene F. and Kenneth R. French, "Common Risk Factors in the Returns of Stocks and Bonds," *Journal of Financial Economics*, 1993, 33 (1), 3–56.
- Fama, Eugene F., "Market efficiency, long-run return, and behavioural finance," Journal of Financial Economics, 1998, 49, 283–306.
- Feenstra, Robert C., "U.S. Exports, 1972-1994, with State Exports and Other U.S. Data," April 1997. National Bureau of Economic Research (Cambridge, MA) Working Paper No. 5990.
- Fisman, David, Ray Fisman, Julia Galef, and Rakesh Khurana, "Estimating the Value of Connections to Vice-President Cheney," May 2006. Working Paper. Columbia University.

- Fisman, Raymond, "Estimating the Value of Political Connections," American Economic Review, September 2001, 91 (4), 1095–1102.
- General Accounting Office, "Report to the Congress of the United States: Impact of Foreign Corrupt Practices Act on U.S. Businesses," 1981. Washington, DC: U.S. General Accounting Office, i-129.
- Gompers, Paul, Joy Ishii, and Andrew Metrick, "Corporate Governance and Equity Prices," *Quarterly Journal of Economics*, February 2003, 118 (1), 107–155.
- Graham, John L., "The Foreign Corrupt Practices Act: A New Perspective," Journal of International Business Studies, 1984, 15, 107–121.
- Guidolin, Massimo and Eliana La Ferrara, "Diamond Are Forever, Wars Are Not: Is Conflict Bad for Private Firms," *American Economic Review*, December 2007, 97 (5), 1978–1992.
- Hines, James R. Jr., "Forbidden Payment: Foreign Bribery and American Business After 1977," September 1995. NBER Working Paper 5226.
- Khotari, S. P. and Jerold B. Warner, Handbook of Corporate Finance. Empirical Corporate Finance, Elsevier, 1997.
- Lambsdorff, Johann Graf, "An Empirical Investigation of Bribery in International Trade," *The European Journal of Development Research*, 1998, *10*, 40–59.
- Loughran, Tim and Jay R. Ritter, "The New Issues Puzzle," The Journal of Finance, March 1995, 50 (1), 23–51.
- Lyon, John D., Brad M. Barber, and Chih-Ling Tsai, "Improved Methods for Tests of Long-Run Abnormal Stocks Returns," *The Journal of Finance*, February 1999, 54 (1), 165–201.
- Olken, Benjamin A. and Patrick Barron, "The Simple Economics of Extorsion: Evidence from Trucking in Aceh," *Journal of Political Economy*, June 2009, 117 (3), 417–452.
- Smith, David B., Howard Stettler, and William Beedles, "An Investigation of the Information Content of Foreign Sensitive Payment Disclosures," *Journal of Accounting and Economics*, 1984, 6, 153–162.
- Svensson, Jakob, "Who Must Pay Bribes and How Much? Evidence from a Cross Section of Firms," The Quarterly Journal of Economics, February 2003, 118 (1), 207–230.

Appendix

A Data: Definitions and Sources

DATA DEFINITIONS AND SOURCES				
VARIABLE	SOURCE	DEFINITION		
Exports	Feenstra (1997)	Industry Exports (4-digit SIC)		
Total Value of Shipments	Bartelsman and Gray (1996)	Value of Industry Shipments		
Export intensity	Feenstra (1997) & Bartelsman and Gray (1996)	Exports over Total Value of Shipments		
Fuels & Electrical Energy	Bartelsman and Gray (1996)	Expenditures on purchased fuels and electrical energy		
Cost of Materials	Bartelsman and Gray (1996)	Total Cost of Materials		
Cost of Energy		Fuels & Electrical Energy over Cost of Materials		
Sales	COMPUSTAT item 12	Net Sales		
Book Value of Equity	COMPUSTAT items 6 (Assets), 181 (Lia- bilities), 10 (Preferred Stock Liquidating Value), 35 (Deferred Taxes & Investment Tax Credit), 79 (Debt-convertible) & 56 (Preferred Stock Redemption Value)	Total Assets less Total Liabilities and Pre- ferred Stock plus Deferred Taxes and Convert- ible Debt. When Preferred Stock is missing, it is replaced with the redemption value of Pre- ferred Stock		
Common Shares Outstanding	COMPUSTAT item 25	Net number of all common shares outstanding at year-end		
Price or Bid/Ask Average	Center for Research in Securities Prices (CRSP)	Closing price or the bid/ask average		
Market Value		Common Shares Outstanding times Price or Bid/Ask Average		
Book to Market		Book Value of Equity over Market Value		
Volume	CRSP	Volume traded, adjusted for splits		
Market Value of Assets		Assets plus Market Value less Book Value of		

TABLE A.1

Equity

TABLE A.1

DATA DEFINITIONS AND SOURCES (CONTINUED)

VARIABLE	SOURCE	DEFINITION
Tobin's q		Market Value of Assets over Assets
Dividends	COMPUSTAT item 21	Total dollar amount of dividends declared on the stock of the company during the year
Dividend Yield		Dividends over Market Value
Adjusted Price -US companies-	CRSP	Stock price close, adjusted for distributions.
Yearly Return		Adjusted Price one-year return rate
Industry Deflator	Bartelsman and Gray (1996)	Price deflator for value of shipments
Real Sales		Sales over Industry Deflator
Real Sales Growth		Real Sales one-year growth rate
US Risk-Free Interest Rate	Federal Reserve	Federal Funds Effective Rate over 365
US Market Return	CRSP	NYSE Total Return on Index
Exchange Rate	Federal Reserve	Nominal Major Currencies Dollar Index
Oil Prices	Dow Jones & Com- pany	Spot Oil Price: West Texas Intermediate
Export Sales	Compustat Business Information Files item 20	Revenue generated by domestically produced goods or services sold outside the domestic country.
Firm Export Intensity		Export Sales over Total Sales
UK Risk-Free Interest Rate	Bank of England	Annual Policy Rate over 365
UK Market Return	Datastream	Daily return on the FTSE All Shares Index
Adjusted Price -UK companies-	Datastream	Stock price, adjusted for changes in capital structure and dividends.
Market Value -UK companies-	Datastream	Share price times the number of ordinary shares in issue.

B Estimates of the Fama French Three Factor Model

	Low Exp.	Med. Exp.	High Exp.
	(1)	(2)	(3)
R^m	0.8979^{**}	1.0144**	1.1504^{**}
	(0.0109)	(0.0079)	(0.0095)
SMB	0.3514^{**}	-0.072**	-0.1466**
	(0.0259)	(0.0187)	(0.0225)
HML	0.0386^{\sim}	-0.1183**	-0.1792**
	(0.0229)	(0.0165)	(0.0199)
Constant	-0.0000	0.0002**	0.0003**
	(0.0001)	(0.0001)	(0.0001)
Observations	1,251	1,251	1,251
R-squared	0.85	0.94	0.93

TABLE B.1

PORTFOLIO REGRESSIONS. FAMA-FRENCH TRHEE-FACTOR MODEL.

Table B.1 shows the estimation of the Fama-French Three-Factor Model for the low, medium and high exports portfolios. The dependent variable is the return on each portfolio. The sample period includes the 1,251 trading days between January 2, 1970 to December 31, 1974. Robust standard errors in parenthesis. Significance levels: \sim : 10%; *: 5%; **: 1%. See the text for further details.

CEMFI WORKING PAPERS

- 0801 David Martinez-Miera and Rafael Repullo: "Does competition reduce the risk of bank failure?".
- 0802 Joan Llull: "The impact of immigration on productivity".
- 0803 Cristina López-Mayán: "Microeconometric analysis of residential water demand".
- 0804 *Javier Mencía and Enrique Sentana:* "Distributional tests in multivariate dynamic models with Normal and Student *t* innovations".
- 0805 *Javier Mencía and Enrique Sentana:* "Multivariate location-scale mixtures of normals and mean-variance-skewness portfolio allocation".
- 0806 Dante Amengual and Enrique Sentana: "A comparison of mean-variance efficiency tests".
- 0807 *Enrique Sentana:* "The econometrics of mean-variance efficiency tests: A survey".
- 0808 Anne Layne-Farrar, Gerard Llobet and A. Jorge Padilla: "Are joint negotiations in standard setting "reasonably necessary"?".
- 0809 Rafael Repullo and Javier Suarez: "The procyclical effects of Basel II".
- 0810 Ildefonso Mendez: "Promoting permanent employment: Lessons from Spain".
- 0811 *Ildefonso Mendez:* "Intergenerational time transfers and internal migration: Accounting for low spatial mobility in Southern Europe".
- 0812 *Francisco Maeso and Ildefonso Mendez:* "The role of partnership status and expectations on the emancipation behaviour of Spanish graduates".
- 0813 Rubén Hernández-Murillo, Gerard Llobet and Roberto Fuentes: "Strategic online-banking adoption".
- 0901 Max Bruche and Javier Suarez: "The macroeconomics of money market freezes".
- 0902 Max Bruche: "Bankruptcy codes, liquidation timing, and debt valuation".
- 0903 Rafael Repullo, Jesús Saurina and Carlos Trucharte: "Mitigating the procyclicality of Basel II".
- 0904 *Manuel Arellano and Stéphane Bonhomme*: "Identifying distributional characteristics in random coefficients panel data models".
- 0905 *Manuel Arellano, Lars Peter Hansen and Enrique Sentana*: "Underidentification?".
- 0906 Stéphane Bonhomme and Ulrich Sauder. "Accounting for unobservables in comparing selective and comprehensive schooling".
- 0907 Roberto Serrano: "On Watson's non-forcing contracts and renegotiation".
- 0908 *Roberto Serrano and Rajiv Vohra*: "Multiplicity of mixed equilibria in mechanisms: a unified approach to exact and approximate implementation".
- 0909 *Roland Pongou and Roberto Serrano*: "A dynamic theory of fidelity networks with an application to the spread of HIV / AIDS".
- 0910 Josep Pijoan-Mas and Virginia Sánchez-Marcos: "Spain is different: Falling trends of inequality".
- 0911 Yusuke Kamishiro and Roberto Serrano: "Equilibrium blocking in large quasilinear economies".
- 0912 *Gabriele Fiorentini and Enrique Sentana:* "Dynamic specification tests for static factor models".

- 0913 Javier Mencía and Enrique Sentana: "Valuation of VIX derivatives".
- 1001 *Gerard Llobet and Javier Suarez:* "Entrepreneurial innovation, patent protection and industry dynamics".
- 1002 Anne Layne-Farrar, Gerard Llobet and A. Jorge Padilla: "An economic take on patent licensing: Understanding the implications of the "first sale patent exhaustion" doctrine.
- 1003 *Max Bruche and Gerard Llobet:* "Walking wounded or living dead? Making banks foreclose bad loans".
- 1004 Francisco Peñaranda and Enrique Sentana: "A Unifying approach to the empirical evaluation of asset pricing models".
- 1005 Javier Suarez: "The Spanish crisis: Background and policy challenges".
- 1006 *Enrique Moral-Benito*: "Panel growth regressions with general predetermined variables: Likelihood-based estimation and Bayesian averaging".
- 1007 *Laura Crespo and Pedro Mira:* "Caregiving to elderly parents and employment status of European mature women".
- 1008 Enrique Moral-Benito: "Model averaging in economics".
- 1009 Samuel Bentolila, Pierre Cahuc, Juan J. Dolado and Thomas Le Barbanchon: "Two-tier labor markets in the Great Recession: France vs. Spain".
- 1010 *Manuel García-Santana and Josep Pijoan-Mas:* "Small Scale Reservation Laws and the misallocation of talent".
- 1101 Javier Díaz-Giménez and Josep Pijoan-Mas: "Flat tax reforms: Investment expensing and progressivity".
- 1102 *Rafael Repullo and Jesús Saurina:* "The countercyclical capital buffer of Basel III: A critical assessment".
- 1103 *Luis García-Álvarez and Richard Luger:* "Dynamic correlations, estimation risk, and portfolio management during the financial crisis".
- 1104 *Alicia Barroso and Gerard Llobet:* "Advertising and consumer awareness of new, differentiated products".
- 1105 Anatoli Segura and Javier Suarez: "Dynamic maturity transformation".
- 1106 Samuel Bentolila, Juan J. Dolado and Juan F. Jimeno: "Reforming an insideroutsider labor market: The Spanish experience".
- 1201 Dante Amengual, Gabriele Fiorentini and Enrique Sentana: "Sequential estimation of shape parameters in multivariate dynamic models".
- 1202 *Rafael Repullo and Javier Suarez:* "The procyclical effects of bank capital regulation".
- 1203 Anne Layne-Farrar, Gerard Llobet and Jorge Padilla: "Payments and participation: The incentives to join cooperative standard setting efforts".
- 1204 *Manuel Garcia-Santana and Roberto Ramos:* "Dissecting the size distribution of establishments across countries".
- 1205 *Rafael Repullo:* "Cyclical adjustment of capital requirements: A simple framework".

- 1206 *Enzo A. Cerletti and Josep Pijoan-Mas:* "Durable goods, borrowing constraints and consumption insurance".
- 1207 Juan José Ganuza and Fernando Gomez: "Optional law for firms and consumers: An economic analysis of opting into the Common European Sales Law".
- 1208 Stéphane Bonhomme and Elena Manresa: "Grouped patterns of heterogeneity in panel data".
- 1209 *Stéphane Bonhomme and Laura Hospido:* "The cycle of earnings inequality: Evidence from Spanish Social Security data".
- 1210 Josep Pijoan-Mas and José-Víctor Ríos-Rull: "Heterogeneity in expected longevities".
- 1211 *Gabriele Fiorentini and Enrique Sentana:* "Tests for serial dependence in static, non-Gaussian factor models".
- 1301 Jorge De la Roca and Diego Puga: "Learning by working in big cities".
- 1302 *Monica Martinez-Bravo:* "The role of local officials in new democracies: Evidence from Indonesia".
- 1303 *Max Bruche and Anatoli Segura:* "Debt maturity and the liquidity of secondary debt markets".
- 1304 Laura Crespo, Borja López-Noval and Pedro Mira: "Compulsory schooling, education and mental health: New evidence from SHARELIFE".
- 1305 Lars Peter Hansen: "Challenges in identifying and measuring systemic risk".
- 1306 Gabriele Fiorentini and Enrique Sentana: "Dynamic specification tests for dynamic factor models".
- 1307 *Diego Puga and Daniel Trefler:* "International trade and institutional change: Medieval Venice's response to globalization".
- 1308 Gilles Duranton and Diego Puga: "The growth of cities".
- 1309 Roberto Ramos: "Banning US foreign bribery: Do US firms win?".